Author's response to reviews

Title: Effectiveness of a Facebook-delivered physical activity intervention for post-partum women: a randomized controlled trial protocol.

Authors:

  Jocelyn S Kernot (Jocelyn.Kernot@unisa.edu.au)
  Timothy S Olds (Timothy.Olds@unisa.edu.au)
  Lucy K Lewis (Lucy.Lewis@unisa.edu.au)
  Carol A Maher (Carol.Maher@unisa.edu.au)

Version: 2 Date: 28 January 2013

Author's response to reviews: see over
29th January 2013

Jocelyn Kernot
Health and Use of Time Group
University of South Australia
GPO Box 2471
Adelaide SA 5001
Australia
Phone: +61 8 8302 1289
Fax: +61 8 8302 2645
Email: Jocelyn.Kernot@unisa.edu.au

Dear Executive Editor,

Please find attached the following formatting changes to our paper titled:

Effectiveness of a Facebook-delivered physical activity intervention for post-partum women: a randomized controlled trial protocol (Study Protocol MS: 1119094111896590)

- The ethics approval and funding proof has been sent as an attachment via email on the 29/01/12
- Please note that this study is not funded by external sources, however we have received a $10,000 University of South Australia Divisional Grant to develop the Facebook app. Data collection will be supported by principal supervisors (Carol Maher) professional development fund and my PhD stipend.
- Copies of Figures one and two were not included in the main manuscript and were submitted under the figures section on the upload page. However, they did appear with the main manuscript when a PDF version was sent via email from BMC website (as if the documents had been combined). I have uploaded the main manuscript and figures separately under the appropriate sections on the upload page again following the website instructions. Please let me know if this continues to be an issue and how I might be able to upload the documents differently.

Please do not hesitate to contact us should you have any further questions. We look forward to hearing from you regarding our paper.

Sincerely,

Jocelyn Kernot